Author's response to reviews

Title: The effectiveness of physical activity monitoring and distance counselling in an occupational health setting - a research protocol for a randomised controlled trial (CoAct)

Authors:

Karita Reijonsaari (karita.reijonsaari@tkk.fi)
Aki Vehtari (aki.vehtari@tkk.fi)
Willem Van Mechelen (w.vanmechelen@vumc.nl)
Timo Aro (timo.aro@ilmarinen.fi)
Simo Taimela (simo.taimela@evalua.fi)

Version: 4 Date: 10 December 2009

Author's response to reviews: see over
Dear editor,

Thank you for your comments. The requested corrections have been made as follows:

1. The title on manuscript and the submission system have been made identical.
2. Trial registration section has been added to the abstract. The updated abstract has been uploaded to the submission system.
3. The reference list has been updated to include all author names for each citation.
4. The tables have been recreated using Tables tool and included in the manuscript.
5. Figures 1 and 2 have been submitted in separate files. Figure numbers have been corrected in the manuscript.
6. Header has been removed from the manuscript.
7. Keywords have been removed from the abstract.
8. Section titled methods has been changes to methods/design in the abstract.
9. Full citation details have been included for references 26 and 38.
10. Appendix has been placed before the acknowledgements section.

Yours sincerely,

Karita Reijonsaari